Author's response to reviews

Title: Between-centre differences and treatment effects in randomized controlled trials: A case study in traumatic brain injury

Authors:

Hester F Lingsma (h.lingsma@erasusmc.nl)
Bob Roozenbeek (b.roozenbeek@erasusmc.nl)
Pablo Perel (Pablo.Perel@lshtm.ac.uk)
Ian Roberts (ian.Roberts@lshtm.ac.uk)
Andrew I Maas (Andrew.maas@uza.be)
Ewout W Steyerberg (e.steyerberg@erasusmc.nl)

Version: 2 Date: 19 April 2011

Author's response to reviews:

As requested, I have added a list of abbreviations and a statement on the ethical approval of the study in the method section